Author’s response to reviews

Title: Efficacy of vitamin C as an adjunct to fluoxetine therapy in pediatric major depressive disorder: a randomized, double-blind, placebo-controlled pilot study

Authors:

- Mostafa Amr (mamr@gnif.org)
- Ahmed El-Mogy (aelmogy@gnif.org)
- Tarek Shams (tshams@gnif.org)
- Karen Vieira (kvieira@gnif.org)
- Shaheen E Lakhan (slakhan@gnif.org)

Version: 2 Date: 12 November 2012

Author’s response to reviews:

Found error in title during submission, corrected.